Author’s response to reviews

Title: Development of a Validation Algorithm for 'Present on Admission' Flagging

Authors:

Terri J Jackson (t.jackson@uq.edu.au)
Jude L Michel (judith.michel@uq.edu.au)
Rosemary Roberts (rroberts@aapt.net.au)
Jennie Shepheard (Jennie.Shepheard@dhs.vic.gov.au)
Diana Cheng (D.Cheng@latrobe.edu.au)
Julie Rust (J.Rust@usyd.edu.au)
Catherine Perry (Catherine.Perry@dhs.vic.gov.au)

Version: 2 Date: 4 March 2009

Author’s response to reviews:

4th March 2009

Anastasios Koutsos, PhD.
Senior Assistant Editor
BMC-series Journals
BioMed Central
Middlesex House
34-42 Cleveland Street
London W1T 4LB

Dear Anastasios,

Further to our recent email correspondence we have attached two new files for
review with our research paper. A copy of the data cleaning algorithm and
instructions on how to interpret and use the algorithm.

If you have any further questions regarding the use of this algorithm please
contact myself or Jude Michel(judith.michel@uq.edu.au).

Yours sincerely,

Terri Jackson, PhD
Associate Professor